Author's response to reviews

Title: Why is it difficult to implement e-health initiatives? A qualitative study.

Authors:

Elizabeth Murray (elizabeth.murray@ucl.ac.uk)
Joanne Burns (burns|2@lsbu.ac.uk)
Carl May (c.r.may@soton.ac.uk)
Tracy Finch (tracy.finch@newcastle.ac.uk)
Catherine O'Donnell (kate.o'donnell@ gla.ac.uk)
Paul Wallace (p.wallace@ucl.ac.uk)
Frances Mair (frances.mair@ gla.ac.uk)

Version: 8 Date: 7 January 2011

Author's response to reviews:

Thank you for giving me the opportunity to proof read this manuscript. I have made the following corrections:

a) standardised the spelling of "normalization", "normalize", "organize" and "organization" so that they are spelt with a "z" throughout (there had been a mix of "s" and "z" spellings).

b) removed the editorial comment left over from an earlier version.

c) removed the underlining of a bracket.

I confirm this version is now ready for publication.